Author's response to reviews

Title: Assessing the efficacy of a modified assertive community-based treatment programme in a developing country.

Authors:

Ulla A Botha (ulla@sun.ac.za)
Liezl Koen (liezlk@sun.ac.za)
John A Joska (john.joska@uct.ac.za)
Linda Hering (Lhering@pgwc.gov.za)
Piet P Oosthuizen (pieto@sun.ac.za)

Version: 3 Date: 20 August 2010

Author's response to reviews:

Dr. Michael Odenwald
Associate Editor
BMC Psychiatry

Dear Sir

Thank you for your very positive feedback re our submission "Assessing the efficacy of a modified assertive community-based treatment programme in a developing country." (MS: 8787922223807720)

In accordance with the reviewer's suggestion we have added a sentence (highlighted in red) to the final paragraph of the discussion, we hope this is what he had in mind.

thanking you in advance

Regards
Ulla Botha
ulla@sun.ac.za